References for a selection of other topics

• Pattern formation on the sphere

P. Chossat, R. Lauterbach & I. Melbourne, Steady-state bifurcation with O(3) symmetry Arch. Rat. Mech. Anal. 113, 313–376 (1991) P.C. Matthews, Transcritical bifurcation with O(3) symmetry. Nonlinearity 16, 1449–1471 (2003)

• Bifurcations from relative equilibria

M. Krupa, Bifurcations of relative equilibria. SIAM J. Appl. Math. 21, 1453–1486 (1990)

• Bifurcations in symmetric maps and symmetric chaos P. Chossat & M. Golubitsky, Iterates of maps with symmetry. SIAM J. Math. Anal. 19, 1259–1270 (1988)

• Bifurcations from periodic orbits with spatio-temporal symmetries J.S.W. Lamb & I. Melbourne, Bifurcation from periodic solutions with spatiotemporal symmetry, in Pattern Formation in Continuous and Coupled Systems eds M. Golubitsky, D. Luss & S.H. Strogatz, 175–192. New York, Springer. (1999) A.M. Rucklidge & M. Silber, Bifurcations of periodic orbits with spatio-temporal symmetries. Nonlinearity 11, 1435–1455 (1998)

• Forced symmetry breaking

B. Sandstede and A. Scheel, Forced symmetry breaking of homoclinic cycles. Nonlinearity 8, 333–365 (1994)

• Coupled cell systems

M. Golubitsky and I.N. Stewart, The Symmetry Perspective. Progress in Mathematics, volume 200. Birkäuser. (2002)

• Cycling chaos

M. Dellnitz, M. Field, M. Golubitsky, A. Hohmann and J. Ma, Cycling chaos. Int. J. Bif. Chaos 5, 1243–1247 (1995) P. Ashwin and A.M. Rucklidge, Cycling chaos: its creation, persistence and loss of stability in a model of nonlinear magnetoconvection. Physica D 122, 134–154 (1998)

• Euclidean symmetry

P. Ashwin, I. Melbourne and M. Nicol, Hypermeander of spirals; local bifurcations and statistical properties. Physica D 156, 364–382 (2001) D. Barkley, Euclidean symmetry and the dynamics of rotating spiral waves. Phys. Rev. Lett. 72 165–167 (1994) B. Sandstede, A. Scheel and C. Wulff, Dynamical behaviour of patterns with Euclidean symmetry, in [23].

References

[1] D. Armbruster, J. Guckenheimer & P. Holmes, Heteroclinic cycles and modulated travelling waves in systems with O(2) symmetry. Physica D 29, (1987) 257–282 [2] P. Ashwin & J.Montaldi, A group theoretic criterion for the existence of robust relative homoclinic trajectories. Math. Proc. Camb. Phil. Soc. 133, 125–141 (2002) [3] I. Bosch Vivancos, P. Chossat & I. Melbourne. New planforms in systems of partial differential equations with Euclidean symmetry. Arch. Ret. Mech. Anal. 131, 199–224 (1995) [4] W. Brannath, Heteroclinic networks on the tetrahedron. Nonlinearity 7, 1367–1384 (1994) [5] P.C. Bressloff, J.D. Cowan, M. Golubitsky & P.J. Thomas, Scalar and pseudoscalar bifurcations motivated by pattern formation on the visual cortex. Nonlinearity 14, 739–775 (2001) [6] F.H. Busse and K.E. Heikes, Convection in a rotating layer: A simple case of turbulence.

Science, 208, 173–175. (1980) [7] E. Buzano & M. Golubitsky, Phil. Trans. Roy. Soc. Lond. A 308, 617–667 (1983) [8] P. Chossat, M. Krupa, I. Melbourne & A. Scheel, Transverse bifurcations of homoclinic cycles.

Physica D 100, 85–100 (1997) [9] P. Chossat, M. Krupa, I. Melbourne & A. Scheel, Magnetic dynamos in rotating convection - a dynamical systems approach. Dyn. of Cont., Discr. and Imp. Syst. 5, 327–340 (1999) [10] P. Chossat and R. Lauterbach, Methods in Equivariant Bifurcations and Dynamical Systems.

World Scientific (2000) [11] C. Cicogna, Symmetry breakdown from bifurcation. Lett. Nuovo Cimento 31, 600–602 (1981) [12] R.M. Clever & F.H. Busse, Nonlinear properties of convection rolls in a horizontal layer rotating about a vertical axis. J. Fluid Mech. 94, 609–627 (1979) [13] G. Dangelmayr, Steady-state mode interactions in the presence of O(2) symmetry. Dyn. Stab.

Syst. 1, 159–185 (1996) [14] J.H.P. Dawes, Pattern selection in oscillatory rotating convection. Physica D 147, 336–351 (2000b) [15] J.H.P. Dawes, P.C. Matthews and A.M. Rucklidge, Reducible actions of D_4 × T^2: superlattice patterns and hidden symmetries. Nonlinearity 16, 615–645 (2003) [16] J.H.P. Dawes, C.M. Postlethwaite and M.R.E. Proctor, Instabilities induced by a weak breaking of a strong spatial resonance. Physica D 191, 1–30 (2004) [17] A. Demircan, S. Scheel & N. Seehafer, Heteroclinic behaviour in rotating Rayleigh-Bénard convection, Eur. Phys. J. B 13, 765–775 (2000) [18] B. Dionne, M. Silber & A.C. Skeldon, Stability results for steady, spatially periodic planforms.

Nonlinearity 10, 321–353 (1997) [19] M.J. Field, Equivariant dynamical systems. Bull. Amer. Math. Soc. 76, 1314–1318 (1970) [20] M.J. Field, Equivariant bifurcation theory and symmetry breaking. J. Dyn. Diff. Eqn. 1, 369– 421 (1989)

[21] M.J. Field and J. Swift, Stationary bifurcation to limit cycles and heteroclinic cyles. Nonlinearity 4, 1001–1043. (1991) [22] M.J. Field, Lectures on bifurcations, dynamics and symmetry. Pitman Research Notes in Mathematics, vol. 356. London: Longman Scientific and Technical (1996) [23] M. Golubitsky, D. Luss & S.H. Strogatz, Pattern Formation in Continuous and Coupled Systems. New York, Springer. (1999) [24] M. Golubitsky, J.W. Swift & E. Knobloch, Symmetries and pattern selection in Rayleigh– Bénard convection, Physica 10 D, 249–276 (1984) [25] M. Golubitsky, I.N. Stewart & D.G. Schaeffer, Singularities and Groups in Bifurcation Theory.

Volume II. Springer, Applied Mathematical Sciences Series 69 (1988).

[26] M. Golubitsky & I.N. Stewart, Hopf bifurcation in the presence of symmetry. Arch. Rat. Mech.

Anal. 87, 107–165 (1985) [27] M. Golubitsky and I.N. Stewart, The Symmetry Perspective. Progress in Mathematics, volume 200. Birkäuser. (2002) [28] J. Guckenheimer & P. Holmes, Nonlinear Oscillations, Dynamical Systems and Bifurcations of Vector Fields. Springer, New York. (1983) [29] J. Guckenheimer and P. Holmes, Structurally stable heteroclinic cycles. Math. Proc. Camb.

Phil. Soc. 103, 189–192 (1988) [30] J. Guckheimer & P. Worfolk, ‘Instant chaos’. Nonlinearity 5, 1211–1222 (1992) [31] J. Hofbauer and K. Sigmund, Evolutionary Games and Population Dynamics, CUP (1998) [32] R.B. Hoyle, Pattern Formation. CUP (2005) [33] K.A. Julien, Strong Spatial Resonance in Convection. PhD thesis, University of Cambridge.

(1991) [34] V. Kirk & M. Silber, Competition between heteroclinic cycles. Nonlinearity 7, 1605–1621 (1994) [35] E. Knobloch & M. Silber, Oscillatory convection in a rotating layer. Physica D 63, 213–232 (1993) [36] M. Krupa, Robust heteroclinic cycles. J. Nonlinear Sci., 7, 129–176 (1997) [37] M. Krupa and I. Melbourne, Asymptotic stability of heteroclinic cycles in systems with symmetry. Ergod. Th. & Dynam. Sys. 15, 121–147 (1995) [38] M. Krupa and I. Melbourne, Asymptotic stability of heteroclinic cycles in systems with symmetry, II. Proc. Roy. Soc. Edinburgh A 134 A, 1177–1197 (2004) [39] G. Küppers & D. Lortz, Transition from laminar convection to thermal turbulence in a rotating fluid layer. J. Fluid Mech. 35, 609–620 (1969) [40] R.M. May and W. Leonard, Nonlinear aspects of competition between three species. SIAM J.

Appl. Math. 29, 243–253 (1975) [41] I. Melbourne, Steady-state bifurcation with Euclidean symmetry. Trans. A.M.S. 351, 1575– 1603 (1999)

[42] I. Melbourne, An example of a non-asymptotically stable attractor. Nonlinearity 4, 835–844 (1991) [43] J. Milnor, On the concept of attractor. Commun. Math. Phys. 99, 177–185 (1985) and Commun. Math. Phys. 102, 517–519 (1985) [44] J. Porter & E. Knobloch, New type of complex dynamics in the 1 : 2 spatial resonance. Physica D 159, 125–154 (2001) [45] M.R.E. Proctor & C.A. Jones, The interaction of two spatially resonant patterns in thermal convection. Part 1: Exact 1:2 resonance. J. Fluid Mech. 188, 301–335 (1988) [46] C.M. Postlethwaite and J.H.P. Dawes, Regular and irregular cycling near a heteroclinic network Nonlinearity 18, 1477–1509 (2005) [47] C.M. Postlethwaite, Stability results for a class of robust homoclinic cycles in R^n. In preparation.

[48] D. Ruelle, Bifurcations in the presence of a symmetry group. Arch. Rat. Mech. Anal. 51, 136–152 (1973) [49] D.H. Sattinger, Group representation theory, bifurcation theory and pattern formation. J.

Funct. Anal. 28, 58-101 (1987) [50] N. Sottocornola, Complete classification of homoclinic cycles in R^4 in the case of a symmetry group G ⊂ SO(4). C. R. Acad. Sci. Paris 334 1–6 (2002) [51] N. Sottocornola, Robust homoclinic cycles in R^4. Nonlinearity 16, 1–24 (2003) [52] A.M. Soward, Bifurcation and stability of finite amplitude convection in a rotating layer.

Physica D 14, 227–241 (1985) [53] J.W. Swift, Hopf bifurcation with the symmetry of the square. Nonlinearity 1, 333-377 (1988) [54] A. Vanderbauwhede, Local bifurcation and Symmetry, Habilitation Thesis, Rijksuniversiteit Gent, (1980). Also Research Notes in Mathematics 75, Pitman, Boston (1982)